Author’s response to reviews

Title: Correlates of Stunting Among Under-five Children in Bangladesh: A Multilevel Approach

Authors:

Papia Sultana (papia.stat@ru.ac.bd)
Mahfuzur Rahman (mdmahfuzur08@gmail.com)
Jahanara Akter (jahanaraju78@gmail.com)

Version: 3 Date: 04 Jul 2019

Author’s response to reviews:

NUTN-D-19-00083R2

Correlates of Stunting Among Under-five Children in Bangladesh: A Multilevel Approach

Dear Editor,

We have revised the paper according to your comment. Revised parts are highlighted.

With best regards,

Papia

Reply to Editor Comments:

Editor:

Please fix minor mistakes as:

- the font of the new paragraph (page 8)

- the punctuation in this sentence: "All computations were done by the authority of the survey and can be found in details in [46] In this study"

- some misspellings [i.e. stunteded (page 10)
-Font of the indicated paragraph has been updated (page 4 highlighted part)
-Updated (page 4, line -8)
-Corrected (page 6, line -14).